Author's response to reviews

Title: A systematic review of integrated working between care homes and health care services.

Authors:

Sue L Davies (s.l.davies@herts.ac.uk)
Claire Goodman (c.goodman@herts.ac.uk)
Frances Bunn (f.bunn@herts.ac.uk)
Christina Victor (Christina.Victor@brunel.ac.uk)
Angela Dickinson (a.m.dickinson@herts.ac.uk)
Steve Iliffe (s.iliffe@ucl.ac.uk)
Heather Gage (h.gage@surrey.ac.uk)
Wendy Martin (Wendy.Martin@brunel.ac.uk)
Katherine Froggatt (s.froggatt@lancaster.ac.uk)

Version: 4 Date: 1 November 2011

Author's response to reviews:

These are formatting changes only